Author's response to reviews

**Title:** Association Between Dietary Fat and Skin Cancer in an Australian Population Using Case-Control and Cohort Study Designs

**Authors:**

Robert H Granger (robert.granger@utas.edu.au)
Leigh C Blizzard (leigh.blizzard@utas.edu.au)
Jayne Fryer (jayne.fryer@workcover.nsw.gov.au)
Terence Dwyer (terry.dwyer@mcri.edu.au)

**Version:** 2  **Date:** 31 January 2006

**Author's response to reviews:**

As per the letter from Iratxe, I was requested to modify my manuscript prior to it being sent out for review. The changes I made were:

1. Included sections in the abstract to conform with the journal (background, methods, results, conclusions), keeping it to less than 350 words. I modified and added, and deleted some sentences within the abstract.
2. In the main text, I renamed the introduction to background, and Materials and Methods to just Methods. I also inserted a phrase into the Competing Interests line at the very end of the article after acknowledgements to read "The authors declare that they have no competing interests".